



# Article The Legal Transition towards a More Circular Electrical and Electronic Equipment Chain—A Case Study of The Netherlands

Ida Mae de Waal 匝

Utrecht Centre for Water, Oceans and Sustainability Law (UCWOSL), Utrecht University, 3512 BK Utrecht, The Netherlands; i.m.dewaal@uu.nl

Abstract: The transition towards a circular economy in the EU requires that the legal framework does not create (legal) barriers, but rather enables or stimulates this transition. At this moment, however, the existing literature argues that the current legal system that governs the life cycle of materials and products—EU chemicals, product and waste legislation—might hamper the transition. This article looks into the legal framework for the transition of the product stream of electrical and electronic equipment (EEE) and more specifically into the general and specific applicable EU chemicals, product and waste legislation, as well as its implementation in the Netherlands. By means of both doctrinal and empirical legal research, this article identifies legal barriers, gaps and incentives in the current legal framework that governs the life cycle of EEE in light of the transition towards a circular economy (CE). To enable or stimulate the CE transition in the EEE chain, it appears that it would be desirable to improve the interaction within this legal system and better align it with the objectives of the CE and life cycle thinking. Moreover, this research shows that a different approach within EU chemicals, product and waste legislation could be beneficial for the CE transition. With regard to EEE, a holistic approach could enable a more circular EEE chain, either within the current sector-approach, or with a more product-specific or material-specific approach within the EU chemicals, product and waste legislation governing EEE.

**Keywords:** circular economy (CE); electrical and electronic equipment (EEE); waste electrical and electronic equipment (WEEE); EU chemicals legislation; EU product legislation; EU waste legislation

## 1. Introduction

The European Union (EU) is in the process of transitioning from a linear economy to a circular economy (CE), which is an economy 'where the value of products, materials and resources is maintained in the economy for as long as possible, and the generation of waste is minimized' [1]. As one of the main building blocks of the EU Green Deal, the transition towards a CE aims to contribute to creating a more sustainable economy for the EU [2]. One of the product groups on which there is a particular focus with regard to the transition towards a CE in the EU, is electrical and electronic equipment (EEE) [3]. EEE is a resource-intensive sector, as EEE consists of a complex and heterogenous mixture of materials and components [2,4-7]. It often contains hazardous substances, as well as critical raw materials (CRMs) [8–10], and other rare and valuable materials. Throughout its life cycle, from the mining of raw materials to the disposal of products, EEE causes negative environmental impacts [11–13], and the used resources and materials can also pose challenges for human health and the environment at the end of life (EoL) stage due to their hazardousness or toxicity [6,10,11,13–15]. Waste EEE (WEEE) is also one of the fastest growing waste streams in the EU [3,10]. Because of growing prosperity [5], rapid innovations and competition within the sector [12,16], and declining product lifetimes [12], the consumption of EEE is expected to keep increasing in the future, simultaneously leading to an increase of WEEE. However, only about 40% of all WEEE is being recycled [3,7], and



Citation: de Waal, I.M. The Legal Transition towards a More Circular Electrical and Electronic Equipment Chain—A Case Study of The Netherlands. *Sustainability* **2023**, *15*, 935. https://doi.org/10.3390/ su15020935

Academic Editor: Antonella Petrillo

Received: 17 October 2022 Revised: 22 November 2022 Accepted: 20 December 2022 Published: 4 January 2023



**Copyright:** © 2023 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the reuse rate is estimated at only 1% of all collected WEEE [8,16]. Significant materials and value are thus being lost to the economy.

The transition towards a more circular EEE chain could not only lead to economic and environmental benefits for the EEE sector itself [17], but could also make a significant contribution to achieving a CE in the EU in general [11]. However, the transition towards a more circular EEE chain has been limited so far [1,7,10,15,18].

The transition towards a CE, including the transition towards a more circular (W)EEE chain, would ideally require, amongst other things, having a legal framework in place that does not hamper the transition, but is conducive to this transition instead. This specifically counts for the legislation that governs the lifecycle of products and materials: EU chemicals, product and waste legislation [19–21]. EEE is governed by sector-specific EU chemicals, product and waste legislation, such as the Directive on the restriction of the use of certain hazardous substances in electrical and electronic equipment (RoHS Directive) [22] and the Directive on waste electrical and electronic equipment (WEEE Directive) [23], as well as more general legislation, such as the Directive establishing a framework for the setting of ecodesign requirements for energy-related products (Ecodesign Directive) [24], and the Waste Framework Directive (WFD) [25]. Extended producer responsibility (EPR) applies to EEE as well, meaning that all those who bring EEE on the European market are responsible for the management of the waste stage of these products [23,25].

Since the first CE Action Plan, the EU has taken several steps to align its legislation with the CE transition, as evidenced by the amendments of the WFD [26], or, specifically with regard to (W)EEE, the amendments of the WEEE Directive [27], and the RoHS Directive [28]. Additionally, following up on its 2020 CE Action Plan, the EC recently announced a proposal for a new Ecodesign for Sustainable Products Regulation [29], as well as reviews of several other legislation governing EEE. Nevertheless, there still seem to be situations in which the transition towards more circular material and product chains is hampered or where there are unexploited opportunities to stimulate the transition through legal measures. This also counts for the transition towards a circular EEE chain. For example, this article shows that EU chemicals legislation poses potential barriers for the recycling of WEEE plastics and the use of secondary plastics, and that the possibility to set circular product requirements is not fully utilized. More in general, the current legal framework that governs the life cycle of materials and products—EU chemicals, product and waste legislation—has been shown to not fully support the transition towards a CE, amongst other things, because of the presumption that there is a lack of coherence between these three areas of law [30]. The idea of having to improve the coherence within the legal framework governing the life cycle of materials and products aligns with the CE and is also in line with and indicated by life cycle thinking, which is key in a CE [19,20].

In light of the transition towards a more circular EEE chain, the aim of this article is to examine legal barriers and incentives with regard to the EU chemicals, product and waste legislation applicable to (W)EEE, in order to determine what can be done to take away these barriers and incentivize the transition. Next to legal doctrinal research, empirical legal research was used, comprising of 14 semi-structured interviews with 16 (W)EEE stakeholders in the Netherlands. The novelty of this article lies in the fact that it combines these two research methods and therewith provides thorough research into the legal transition towards a more circular EEE chain, from both a practical and theoretical perspective.

The article is structured as follows. Section 2 sets out the methodology of the literature and interview study and the choice for examining the Netherlands. Section 3 focuses on the EU chemicals, product and waste legislation governing EEE, both in general and in relation to the transition towards a CE. Section 4 contains the results of both studies and sets out the identified legal barriers and (lack of) incentives for a more circular EEE chain. Section 5 analyzes the identified legal barriers and (lack of) incentives, also in light of future developments. Finally, Section 6 contains the conclusion.

## 2. Materials and Methods

The aim of this research has been to broaden the discussion on legal barriers and incentives for the CE transition in the EEE chain beyond a theoretical point of view. Therefore, the choice was made to combine a literature study with conducting interviews with (W)EEE stakeholders along the EEE product chain in the Netherlands.

The literature study aimed to identify legal barriers and incentives for the CE transition in the (W)EEE chain, by examining both (legal) literature, policy documents and case law that make mention of (W)EEE and the CE transition, or related and preceding concepts such as life cycle thinking, lifetime extension or resource efficiency. This literature was found by searching online databases (Scopus, Worldcat, Google Scholar and Legal Intelligence), the online databases of the EC, the Council of the European Union and the European Parliament as well as the databases of the CJEU and the Dutch judiciary. In addition, the snowballing method was used to select further relevant literature.

The main aim of the interview study was to, in light of the transition towards a circular (W)EEE chain, identify possible legal barriers, solutions and (lack of) incentives regarding EU chemicals, product and waste legislation applicable to (W)EEE as well as to validate identified legal barriers, gaps, solutions and incentives mentioned in literature and policy documents. Based on purposive and snowball sampling [31,32], 16 stakeholders were interviewed in 14 semi-structured interviews between March–May 2022 [33–35]: 6 with producer (organizations), 2 with refurbishers, 4 with treatment facilities, and 2 with governmental organizations. The interviews were semi-structured to allow interviewees to introduce new topics and to have the flexibility to discuss certain topics more in-depth. A set list of questions was used for each interview, while leaving room to deviate from this list or to elaborate on particular topics (see Appendix A for the questionnaire) [33]. The transcripts of the interviews were de-identified, coded using NVivo and analysed together with the results of the literature study using a theoretical thematic analysis [36]. As only stakeholders in the Netherlands were interviewed, it is noted that the interview study is not representative for all EU MSs. However, the outcome could be relevant to other MSs as well, as it gives insight into the relation between identified barriers and (lack of) incentives in literature and the experienced barriers and (lack of) incentives in practice as well as into the implementation of EU chemicals, product and waste legislation applicable to (W)EEE.

The decision was made to conduct interviews with stakeholders in the Netherlands. In general, from recycling statistics it appears that the Netherlands is among the best performing half of the MSs in Europe [37,38]. However, with regard to the collection and recycling of WEEE, based on the (most recent) statistics of 2018 and 2019, the Netherlands currently belongs to the worse performing half of the MSs, with a collection rate of 49%, and a recycling percentage of 41% [39–41]. While the Netherlands did achieve the minimum recovery and recycling targets for all main categories, it has failed to reach the overall collection target of 65% that has been in place since 2019 [41–43]. Figure 1 clearly shows an increase of the distance to the target of 65% and, compared to 2020, the amount of collected WEEE has decreased by 7% in 2021. It also becomes apparent that PV panels have an important impact on the collection rate, as their share of the three year average POM is almost one third while the collection of PV panels is next to nothing [43]. In short, Figure 1 shows that over half of all WEEE is currently not being recycled nor recovered. Unfortunately, the interviews were conducted before the results over 2021 were published, so the interviewees could not be asked what, in their opinion, caused the deteriorated results, therewith representing a gap for further research.

Next to the implementation of the EU chemicals, product and waste legislation governing EEE (see Appendix B), the Netherlands has also taken additional steps. With regard to the EPR for WEEE, a so-called 'generally binding agreement' (*algemeen verbindend verklaring* (AVV)) is applicable in the Netherlands. This AVV formalizes an EPR initiative from the sector, which means that all producers and importers of a product must contribute to the EPR scheme that is laid down in the AVV. In the case of EEE, the AVV requires all producers and importers to join the producer responsibility organization (PRO) 'Foundation OPEN' *Stichting OPEN*), and to contribute a waste management fee to this PRO. Additionally, the voluntary CENELEC standards (previously WEEELABEX) are mandatory in the Netherlands, meaning that waste treatment facilities have to comply with those [44]. Nevertheless, as stated before, the collection and recycling results are disappointing and show a growing failure to reach the target of 65%.

On balance, the Netherlands poses as an interesting case to examine which existing approaches were or still are successful, but at the same time to look at ways in which the existing framework can or even should be modified in order to realize a more circular EEE chain.



Figure 1. Achieved results in the Netherlands in 2021 and 2020 [42,43].

## 3. EU Chemicals, Product and Waste Legislation Governing WEEE in Light of the CE

Under EU law, (W)EEE is governed by both general and sector-specific EU chemicals, product and waste legislation, which together can be said to govern the whole life cycle of EEE [19–21,30]. The main legal acts within these three areas are the RoHS Directive, the REACH Regulation [45], and the Classification, Labelling and Packaging Regulation (CLP Regulation) [46]; the Ecodesign Directive and the Energy Labelling Framework Regulation [47]; and the WEEE Directive, the WFD and Waste Shipment Regulation (WSR) [48]. Where the RoHS Directive and the WEEE Directive are sector-specific legal acts, the other acts can be considered key legal acts in their respective areas of law and are, to varying degrees, considered to be comprehensive legal frameworks.

In a CE, life cycle thinking is considered key as it enables optimization of the life cycle of materials or products and further enables the assessment of their overall environmental impact [49–51]. Therefore, next to providing some additional insight into the key provisions of the legislation, the following overview shows whether both the CE and life cycle thinking are taken into account in the abovementioned legislation. It also includes the interaction between the legislation governing the life cycle, which is also relevant in light of both the CE and life cycle thinking [30].

Table 1 contains a concise overview of the main legal acts governing EEE, including their objectives and key provisions, as well as a brief description of the extent to which the CE or life cycle thinking are taken into account. Appendix B contains a brief overview of the implementation of these legal acts in the Netherlands.

Focuses on: Use of Chemicals (C), and/or Objective of the Description, including any The Taking into Account of the Legal Act Product Stage (P) and/or Legislation **Key Provisions CE or Life Cycle Thinking** Waste Stage (W) Restricts the use of ten Life cycle thinking is Regulates the use of hazardous substances, reflected in the objective, and hazardous substances in including flame is explicitly mentioned to EEE to contribute to the retardants, heavy metals apply when assessing the protection of human and plasticizers (art. 4 (1) impacts of exemptions (art. 5, RoHS C (W) health and the jo. Annex II). Annex V). Directive environment, including Contributes to The interaction with the environmentally sound improving flow of waste stage is to be taken recovery and disposal of information on into account in the review WEEE (art. 1). substances in EEE. and amendments of restricted substances (art. 6). Aims to ensure a high level of protection of Underlines the importance of human health and the taking into consideration the environment from adverse Sets up a system for whole life cycle of substances, REACH effects of chemical registration, evaluation, which becomes apparent with С Regulation substances, as well as to authorization and restriction respect to the preparation of chemical safety reports and support the functioning of of chemical substances. the internal market and assessments (art. 3.37, art. 18(4)(a), stimulate competitiveness Annex I, Annex II, Annex XII). and innovation (art. 1 (1)). Aims to ensure a high Establishes the system of level of protection of classification, labelling and CLP human health and the Does not mention nor reflect the С packaging of chemical Regulation environment, and the free CE or life cycle thinking. substances and mixtures in the movement of substances EU. (art. 1(1)). Ecodesign means that the environmental aspects are integrated into the product Establishes a framework design to 'improve the for setting ecodesign environmental performance requirements for of the product throughout its Aims to ensure the free energy-related products whole life cycle' (art. 2 (23)). movement of in the EU (art. 1 (1) and Ecodesign Ecodesign requirements can P (C) (W) energy-related products (2)).Directive include aspects that within the internal market Ecodesign requirements contribute to the CE, such as (art. 1(1)). are implemented resource efficiency [52,53]. through product-specific Implementing regulations implementing that also take CE aspects into regulations. account, albeit to a limited extent, are in force since 2019 [54]. Sets the framework for Aims to ensure the the labelling of functioning of the internal Current product-specific energy energy-related products, Energy market and high level of labelling regulations for EEE do regarding energy Labelling environmental and not provide information on efficiency, consumption Framework Р consumer protection, as circularity aspects, instead the of energy and other Regulation well as to reduce negative predominant focus is on energy resources during use (art. environmental impacts of efficiency. 1(1).products [55]. Complements ecodesign requirements.

**Table 1.** Overview of EU chemicals, product and waste legislation governing EEE in light of the CE transition.

Legal Act	Focuses on: Use of Chemicals (C), and/or Product Stage (P) and/or Waste Stage (W)	Objective of the Legislation	Description, including any Key Provisions	The Taking into Account of the CE or Life Cycle Thinking
WEEE Directive	W (P)	Aims to protect human health and the environment and contribute to sustainable development by preventing and improving the waste management of WEEE, by reducing the impacts of resource use, and by improving the efficiency of resource use (art. 1).	<ul> <li>Requires MSs to take appropriate measures for ensuring that ecodesign requirements facilitating reuse and treatment of WEEE are applied (art. 4).</li> <li>Requires separate collection as well as proper treatment of WEEE (art. 5, art. 8).</li> <li>Sets minimum collection target to be achieved annually in each MS of 65% of the average weight of EEE placed on the market (PoM) in the three preceding years in the Member State concerned or 85% of WEEE generated on the territory of that Member State (art. 7).</li> <li>Contains specific recovery and recycling targets for each EEE category (Annex V).</li> <li>Contains provisions on the setting and development of standards for the treatment of WEEE, which has resulted in the CENELEC standards (art. 8 (5)) [56,57].</li> </ul>	<ul> <li>Life cycle thinking is reflected in its objective, in which the interaction with and supplementation of EU waste management legislation, such as the WFD, is also mentioned (art. 1).</li> <li>Multiple provisions reflect CE and life cycle aspects, e.g., the establishment of EPR [50]; the fact that MSs are to encourage the cooperation between producers and recyclers with a view to improve product design for the end-of-life treatment (art. 4) [58]; MSs are to ensure that collecting and transporting WEEE provide for optimal conditions for the preparation of reuse and recycling (art. 6); and there are requirements on proper treatment for certain materials and components (art. 8 jo. Annex VII and VIII).</li> </ul>
WFD	W (C) (P)	Aims to protect human health and the environment by preventing or reducing the (adverse impacts of the) generation of waste; by reducing the overall impacts of resource use; and by improving the efficiency of resource use (art. 1).	Provides the general framework for the waste management in the EU, including the waste hierarchy (art. 4); by-products and end-of-waste criteria (art. 5, art. 6); EPR (art. 8, art. 8a); and targets for the preparation for reuse and recycling of waste (art. 11).	<ul> <li>Explicitly mentions that its objective is 'crucial for the transition to a circular economy' (art. 1).</li> <li>Multiple provisions underline the importance of looking at the life cycle of materials and products as a whole, or are based on life cycle thinking, such as the waste hierarchy or EPR (e.g., art. 4, art. 6, art. 8, art. 9).</li> <li>Regularly mentions the interaction between the WFD and EU chemicals and product legislation (art. 6, art. 9).</li> </ul>
WSR	W	Aims to reduce risks for human health and protect the environment [59].	Establishes procedures and control regimes for shipments of waste between MSs, imported from or exported to third countries (art. 1).	Neither the CE nor life cycle aspects are explicitly mentioned in this legal act.

Table 1. Cont.

Table 1 shows that some of the legal acts take into account the CE or life cycle thinking, or refer to the interaction between the different areas of law or legal acts. For example, the objective of the WFD explicitly refers to the transition to a CE and the WEEE Directive requires that MSs take appropriate measures to ensure that ecodesign requirements are applied that facilitate reuse and treatment of WEEE. In addition, the CE, life cycle thinking or the interaction between the legal acts are often referred to in the recitals in the preamble of the legal acts. Recitals are not legally binding, but provide the reasons on which the legal acts are based and serve as an interpretative tool [60]. For example, recital 2 of the 2017 revision of the RoHS Directive shows that the Directive focuses on the secondary market operations of EEE, such as the availability of spare parts, repair and reuse and therewith to contribute to the CE transition [28,61]. Recital 36 of the Energy Labelling Framework Regulation states, for example, that the analysis for further product groups for which an energy label could be established, should also look at supplementary information to be included, such as durability or environmental performance, in coherence with the objective to promote a CE [47]. Recitals 6 and 11 of the 2012 recast of the WEEE Directive already underlined that the first priority in contributing to sustainable production and consumption is to prevent WEEE, and that the Directive seeks to improve 'the environmental performance of all operators involved in the life cycle of EEE' [23], whereas recital 1 of the 2018 amendment of the WEEE Directive states that waste management should be improved with a view to, amongst other things, 'promoting the principles of the circular economy' [27]. With regard to the interaction, several of the recitals of the RoHS Directive mention that it supplements the general EU waste management legislation, interacts with the Ecodesign Directive, and also refers several times to the interaction within EU chemicals legislation, especially referring to the REACH Regulation [22]. In addition, the recitals of the WEEE Directive mention several times its interaction with the Ecodesign Directive [23], and the recitals of the Ecodesign Directive and WFD explicitly mention the synergies and interaction between the abovementioned legal acts and areas of law [24,26].

In short, in addition to the legal acts themselves that take into account CE objectives, life cycle thinking, or the interaction among the legal acts, the recitals of several legal acts also mention these aspects. In general, however, the legislation is still to a large extent 'linear' as it predates the CE transition, but this may change in the near future as multiple reviews of existing legislation as well as proposals for new legislation have been introduced (see further Section 5.1).

# 4. Results

The literature study and interview study identified a wide variety of barriers and incentives for the transition towards a more circular EEE chain, including many legal barriers and (lack of) incentives that relate to the EU chemicals, product and waste legislation governing EEE. These barriers and incentives were identified at three levels: within EU legislation, at the interface between EU legislation and national legislation and at the national (Dutch) level.

Although the article is limited to discussing these legal barriers and (lack of) incentives related to EU chemicals, product and waste legislation governing EEE, multiple other issues and incentives were identified, each representing an area requiring further research, for example with regard to:

- different legislation, for example consumer legislation and green public procurement [18,62–64];
- the behavior, attitude and awareness of consumers and producers, such as the lack of awareness and attitude of consumers with regard to recycling WEEE and their willingness to buy refurbished products [6,15,52,62,63,65–68];
- the collection of WEEE [6,66,69];
- the cooperation, transparency and flow of information within the supply chain [6,12,62,70];
- data safety and IP [12,71];
- financial and economic aspects, such as deposit refund schemes [6,8,63,69,72,73];

## information and knowledge on circular (W)EEE management [5,65,66,72,74];

The identified legal barriers and (lack of) incentives regarding EU chemicals, product and waste legislation often relate to existing legislation, and include barriers with regard to existing provisions or (lack of) incentives related to these acts. These will be set out below. For the sake of completeness, it should be noted, however, that the following overview does not claim to be a comprehensive overview of all existing legal barriers and (absent) legal incentives.

## 4.1. EU Chemicals Legislation

It is important that WEEE is disposed of and treated properly, as (W)EEE can contain various hazardous substances, such as flame retardants, PCBs and PBDEs, that could pose a threat to human health and the environment [11]. However, from a CE perspective WEEE is not only considered waste, but also a resource [69]. Therefore, the presence of hazardous substances as well as the legislation governing them can form a barrier for the refurbishment and recycling of WEEE. Two interviewees remarked that refurbished products that are placed on the market must comply with the most recent EU chemicals legislation, i.e., the RoHS Directive [75]. According to an interviewed producer, this hampers the refurbishment of EEE, as they do not have the information that is required to check if these refurbished products and components comply with this new legislation [18]. The legacy substances problem was also mentioned [21,76–78], meaning that despite the regulation of the use of hazardous substances, they continue to be present in waste streams as it not only takes a long time to phase out these hazardous substances [69], but products containing hazardous substances can still enter the EU market due to online shopping [11]. In the literature, it has furthermore been noted that, despite the fact that the REACH Regulation requires that the whole life cycle of substances should be taken into account with regard to chemical safety reports and assessments, the use and end-of-life stages are not considered detailed enough [64].

Issues with regard to hazardous substances present in WEEE streams often concern the presence of hazardous substances in plastics. In the literature, chemicals legislation, more specifically the REACH Regulation and RoHS Regulation but also the POPs Regulation, were named as a common barrier for recovering and recycling plastics in EEE, as well as for using recycled plastics in new EEE [12]. This aligns with interview statements by waste treatment facilities and producers, who stated that the legislation on flame retardants hampers the recycling of WEEE plastics and the use of secondary plastics. Not only because of the legacy substances problem, but also because they will not be able to measure the refined concentrations of flame retardants that are being proposed as new thresholds for EEE plastics [79]. Furthermore, the reuse, refurbishment and recycling of WEEE, as well as the use of secondary materials in EEE have been said, in both literature and interview, to be hampered, as the rules in EU chemicals legislation, more specifically the REACH regulation and ROHS Directive, cause uncertainties and administrative burdens [18].

## 4.2. EU Product Legislation

The Ecodesign Directive is often mentioned when it comes to legislation that could stimulate the transition towards a circular economy as well as limit the environmental impact of products [64]. It provides the possibility to set ecodesign requirements that could contribute to the creation of products that already take into account the subsequent life cycle stages [24,68], as well as contribute to the lifetime extension [80,81], durability [82], and resource efficiency of products [64,82,83]. Ecodesign requirements can, moreover, serve as a 'push' instrument for producers and can create a level playing field in the EU [64,84].

Only a few of all possible ecodesign requirements to contribute to these objectives include requirements on easy disassembly (e.g., time thresholds for disassembly [84–87], standardization of fastenings [52], mandatory dismantlability of key components [88]), on material efficiency [88], on recycled content [12,52,64,72,89], on promoting repairability (e.g., availability of spare parts or software for a specified number of years) [72,80], or on

material use (e.g., ban of (hazardous) materials that negatively affect recycling or reuse, concentrated use of priority materials, use of mono-materials) [52,64]. [52,67] Interviewees added, inter alia, the possibilities of setting requirements on the number of allowed plastic types to facilitate the waste treatment process [52], or requirements that accommodate color separation in the waste treatment process. They also remarked that a requirement on recycled content could boost demand for recycled materials, especially for plastics.

Ecodesign requirements could, moreover, contribute to improve the flow of information within the supply chain [72], for example, by setting information requirements on resource efficiency [64], the presence of hazardous substances or CRMs [52,71,89], or on disassembly, disposal or recycling [52]. This could inform consumers on resource aspects, help end-users to correctly dispose their products, and help improve the efficiency and safety of recycling and recovery processes. However, such information has to be easily accessible for the relevant actors. According to some interviewees, an attached QR code could be a possibility for consumers and repairers, but all but one of the waste treatment operators stated that this would not work, as scanning and looking up information per product does not fit in the current waste treatment process. As an alternative, it was suggested to embed information on the product itself, for example, by means of color marking or logos [52].

Some of the abovementioned suggestions are already implemented in ecodesign implementing measures for a few product groups, such as requirements on the reusability of components and on compulsory available software [54,72,90,91]. In general, however, limited use has been made of the possibility to include CE aspects in ecodesign requirements [30]. A shift to improve the focus of ecodesign requirements from energy efficiency to resource efficiency and lifetime extension would be desirable to stimulate the CE transition [52,88]; also according to six interviewees. This does require, however, that the methodology for ecodesign of energy-related products - the assessment for ecodesign requirements on a specific product group [92] - is adjusted to ensure that resource efficiency and durability aspects can actually be implemented in ecodesign requirements [52,64,88,93]. In addition, it is important that there are adequate standards and methods in order to prove compliance, to monitor and to enforce these ecodesign requirements [52,54,67,83,88,89,94]. Nevertheless, it has been pointed out in literature and by an interviewee that it is difficult to check if products meet product requirements, especially as products are mostly manufactured outside of Europe [16,52,93]. Lastly, it has been argued that too many and too detailed ecodesign requirements might actually hamper innovation, which is undesirable for the CE transition [95,96]. This aligns with the comment of an interviewed producer of circular products, who is against more stringent product requirements for this reason.

Next to the possibilities of the Ecodesign Directive, additional options were mentioned in the literature to increase, amongst other things, the durability and repairability of products. Some options concerned the potential of the EU energy label or the voluntary Ecolabel. Similar to how these product policy instruments currently address energy efficiency, both labels could also address CE criteria and therewith serve as 'pull' instruments for both consumers and producers [63,64,81,84]. The EU Energy label could, for example, provide information on durability aspects, such as an average expected product lifetime, and repairability [62,85], whereas the Ecolabel could address resource efficiency [63,84]. Another mentioned option for the EU to promote durability is to enter into voluntary agreements on durability with the (W)EEE industry [62].

#### 4.3. EU Waste Legislation

The WEEE Directive has as its objective to prevent the generation and disposal of WEEE and to retrieve materials contained in EEE. From previous research, it appeared that the WEEE Directive, including its requirements, targets and standards, is considered an enabler for implementing circular business models in the EEE chain [12]. At the same time, however, the WEEE Directive is criticized for encouraging recycling at the expense of higher options in the waste hierarchy, such as reuse and repair [6], and for hampering the transition to a more circular EEE chain [12]. It is argued that the objectives of the

WEEE Directive are not reflected in its provisions, as targets are not consistent with these objectives [69]. The Directive does not contain specific incentives for (preparation for) reuse or for specific materials, only for collection and recycling [73], despite the fact that the Directive provides a possibility for setting a target for the preparation of reuse in Article 11(6) and that targets for specific materials were to be considered according to recital 7 of the 2018 amendment [23,27]. The current structure of the WEEE Directive is also said to lead to actors putting all their effort into meeting the collection and recycling targets instead of focusing on higher waste hierarchy steps [73], which is supported by four interviewees criticizing the Dutch PRO Stichting OPEN for only fixating on achieving the 65% target and using its position in such a way that reuse, repair and refurbishment are hampered in favor of material recycling by shredding; all in order for them to reach that target.

Looking at the specific instruments, in five interviews, the current structure of the target itself was perceived as an obstacle to a more circular EEE chain, as aiming for higher levels of the waste hierarchy negatively influences the amount of waste that can be collected and therefore influences the achievement of the 65% target. However, another interviewee mentioned that reuse actually can be included in the recycling and recovery target, but that the registration—which takes place in the waste stage—is the problem [74]. Two interviewees remarked that the criticism of the target also seems to be used as an excuse for not achieving the target itself. As stated before, the Netherlands did not achieve the collection target of the WEEE Directive (see Section 2).

The literature also criticized the references of the existing targets: in general for being mass-based, therewith neglecting the (ecological) value and scarcity of materials in EEE [69,97], and specifically the 65% collection target because its 'Put on Market' reference does not properly reflect the amount of WEEE that is generated or available for collection [74]. One waste treatment operator acknowledged that the recycling of CRMs makes no difference if targets remain weight-related, given their low weight share. Furthermore, three interviewees were critical of the collection target references, in particular, as the target wrongly assumes that EEE PoM is discarded after 3 years.

Moreover, the WEEE Directive is argued to fail to incentivize producers to focus more on ecodesign, as Article 4 does not guarantee that the design of EEE is indeed facilitating reuse, dismantling and recovery of WEEE [16], and because producers cannot get their product or material back at its end-of-life [89]. In addition, the information obligation in Article 15 is not considered effective [21,85,94]. A lack of legislation for access to discarded products for preparation for reuse [16], and for preventing leakage flows as well as a lack of enforcement in preventing leakage flows were also mentioned [18]. The fact that the implementation of the WEEE Directive leads to different rules and requirements among MSs was also perceived as a barrier [12,18].

To solve the issues regarding the targets in the WEEE Directive, it has been suggested to set more stringent targets [98], introduce new targets, such as for reuse and preparation for reuse [6,99], or introduce qualitative targets, focusing on certain materials or the quality of collected products [97,100,101]. According to two interviewees, such targets are possible, but do require that the need for achieving such targets is already taken into account at the product stage. Specifically with regard to the 65% collection target, it was suggested to change the reference from WEEE placed on the market to WEEE that is actually generated or available for collection [74]. An adjustment of this target was also considered by two interviewees, while the registration of reuse and export for reuse were also believed to be important for contributing to the achievement of the 65% collection target.

Other incentives that were reported with regard to the WEEE Directive range from the revision of the Directive to more directly address the period before EEE is discarded [6], to introduce mandatory testing for repairability or reuse [16], to revise article 6(2) to ensure that the reusability of EEE is safeguarded during its collection [99], and to introduce a landfill ban [98], and a mandatory certification scheme for recyclers [102]. To improve the actual collection in the Netherlands, another incentive that already has been proposed is the so-called legal handover obligation (*wettelijke afgifteplicht*), requiring all parties who receive

WEEE from consumers to hand it over to CENELEC certified treatment facilities [103], since WEEE still ends up in non-compliant schemes despite the fact that CENELEC standards are mandatory in the Netherlands [74].

As EPR applies to EEE, it plays an important role in achieving the objectives and targets from the WEEE Directive. Despite the fact that EPR is said to have had a positive effect on the collection and recycling of WEEE [89,101,104], as stated before the current collection targets were in 2019 only met by three MSs, not including the Netherlands [105]. In the Netherlands, for example, approximately half of all generated WEEE is not disposed of properly, is exported for reuse or non-compliantly recycled [74]. More in general, the instrument of EPR is criticized in the literature for not achieving the desired results. EPR is intended to positively influence the design and production of EEE as well as to take into account the whole life cycle and thereby strengthen the reuse of EEE as well as the prevention, recycling and other recovery of WEEE [52,53,89]. However, the quality of the recycling is disputed and the effect of EPR on the design of products is minimal. EPR is therefore said to be not sufficient for guaranteeing sustainable waste management [89,101,104,106].

Nine interviewees were critical of the instrument of EPR, with some of them criticizing the EPR governance in the Netherlands in particular. According to three interviewees, EPR has potential, but it is very complex to achieve its objectives. The limited insight and influence of producers on other actors in the still linear market was mentioned as an explanation [107]. While this made one producer wonder if the responsibility within EPR is properly distributed, another interviewee was critical of the role of producers because of their conflicting interests, as they still benefit from a linear economic model. Other interviewees, including one producer of circular products, stated that the current EPR system hinders producers to act more circularly as EPR also applies to producers who take back their own products or take other circular measures such as preparation for reuse. Moreover, barriers can be related to the way in which the MSs have set up their national EPR schemes [12]. In the Netherlands, it becomes apparent that in 2021, the collection of WEEE has substantially decreased and that the distance to the 65% target has increased (Section 2), despite the establishment of PRO Stichting OPEN. In addition, interviewees were critical of the Dutch set up as a pay-as-you-go system (*omslagstelsel*) for not taking provisions for future challenges, and of the way Stichting OPEN uses their position in the market, for example, with regard to contracts.

Suggestions for improving the EPR framework were also identified. Among other things, stronger enforcement [12], introducing modulated fees - adjusted to and therewith stimulating the circular design of products and favoring circular producers [12,63,76,106]-, and extending EPR by including take-back requirements [98] were mentioned in the literature. Interviewees further suggested to renew the governance structure of EPR by including people who monitor and supervise the contribution of the EPR systems to the CE or to think about an EPR design that includes more than just buying out producers' responsibility for the end-of-life stage [106]. With regard to the functioning and position of the PRO Stichting OPEN, interviewees mentioned the idea to only use Stichting OPEN's structure for critical waste streams instead of all WEEE and leave other waste streams to the market itself, or to split up the current AVV for all WEEE, because of the differences between all WEEE products. Moreover, it was suggested in both literature and the interviews that it should be clarified that in case of (preparation for) reuse, repair, or refurbishment, one should not be considered to be a producer within the meaning of EU legislation, including EPR [62]. Modulated fees were, in contrast to the literature, only considered positive by one interviewee, while three other interviewees were critical about its potential.

The observation that the sector-specific waste legislation, i.e., the WEEE Directive, focuses too little on higher R-strategies, such as reuse, also applies to the waste legislation in general, including the WFD. In the literature, it is stated that there is too little focus and guidance on higher steps in the waste hierarchy [6,108], and that, in general, the focus on recycling within waste law forms a barrier for longer lifetime and repairability

of products [63]. This was also recognized by one interviewee, who remarked that the precedence of reuse over recycling is not well anchored in legislation. Another issue relating to the WFD, that came up several times in the interviews, was the influence of the waste status on reuse of EEE. Five interviewees indicated that it is difficult to transition something that has been classified as waste to a product, and that this hampers the reuse or refurbishing of products. One interviewee also pointed out that the waste status advantages virgin raw materials and resources over secondary ones, as the former do not fall under the as stringent perceived waste legislation and WSR (see below). In contrast, however, two of the interviewees indicated that the waste status is not a problem for the reuse of components, with one giving the treatment according to the CENELEC standards as an explanation. Two other interviewees, both working for a government organization, were more critical of the issues regarding the waste status and wondered whether it is truly the waste status itself or whether it is something else, such as the image of waste, that hampers the CE transition.

Incentives often focused on the waste status as well. For example, in the literature, it has been suggested to clearly differentiate between 'reuse to prevent waste' and 'reuse that follows material recovery processes' [11]. One of the interviewees stated that the focus should not be on the definition of waste itself, but rather on solving actual issues in legislation that hamper CE activities when something is being classified as waste, such as an exemption for the permit requirements for recovery or waste disposal for circular repair centers. Further, creating more end-of-waste criteria was mentioned as a priority incentive by the same interviewee.

In the literature and the interviews, multiple barriers were also identified regarding the WSR and the export of (W)EEE. In the literature, cross-border shipments of used EEE to refurbishment or recycling sites are said to be hampered by unclear and overlapping legislation [18]. This corresponds to the experiences of an interviewed producer, who experiences limitations when shipping products for refurbishment, as they are constantly dealing with questions on whether it is possible to ship products that are suitable for refurbishment or repair to designated refurbishment sites abroad. Another barrier for optimal recycling of WEEE that is identified in the literature is that the WSR is complex and leads to lengthy administrative processes and high administrative costs, therewith making transactions of WEEE for repair or recycling more difficult [18,89]. This has been endorsed by three interviewed waste treatment facilities. Especially, the notification procedure of the WSR and the Euralcodes are, respectively, considered burdensome and not working properly. Another barrier that is regularly mentioned in the literature relates to a lack of enforcement and monitoring to restrict, inter alia, informal or illegal export of 'sham reuse' to developing countries [14,15,18,66,70], despite the fact that the Euralcodes and CENELEC standards should contribute to preventing this. While interviewees also mentioned a lack of enforcement, one government organization official stressed, however, that it is difficult for authorities to check if exports contain functioning EEE instead of WEEE. In addition, one interviewee argued that there is a lack of proper regulation for the trade in second-hand EEE.

### 5. Discussion

The previous section has shown that identified legal barriers and incentives in the literature largely correspond to the legal barriers and incentives that were identified by stakeholders in the Netherlands. Although the opinions of interviewees differed with regard to the implementation of information requirements and the possibility of modulated fees in EPR, in almost all other cases the barriers and incentives identified in the literature were endorsed and often supplemented by interviewees based on their experiences in practice. On top of that, interviewees also regularly emphasized the importance of general aspects of legislation, i.e., the necessity of legislation being enforceable, in line with practice and not too complex. Looking at the overview of the legal barriers and (lack of) incentives above, three main themes could be observed.

First, several of the identified barriers and incentives seem to be related to aspects linked to the implementation of the examined legislation, including challenges related to measurement methods and standards, its enforcement and more general aspects, such as its complexity. In both the literature and the interviews, it was stated that current measurement methods should be adjusted, that appropriate methods and standards should be developed and that the measurability of CE aspects should be improved. In addition, calls for the usefulness and necessity of a level playing field in the EU were made regularly [89], in particular, to reduce or prevent regulatory disparities between MSs. The lack of enforcement of legislation is also clearly a problem that affects a large part of the examined legislation. The interviews revealed that this can be due to the legislation itself, for example, the difficulty in checking if products comply with certain product requirements. However, according to interviewees in the Netherlands, it seems to be particularly due to more practical and non-legal aspects, such as a lack of manpower, money or knowledge, either on the part of the competent authority, or on the part of other market actors, such as a lack of awareness among consumers. More in general, across all legal acts, the legislation seems to be leading to uncertainties and administrative burdens or is considered to be too complex, which hampers the reuse, refurbishment and recycling WEEE.

Second, the identified barriers and incentives reveal that the interaction between the three legal areas can lead to (potential) issues and therefore could or should be improved. This becomes apparent in the aforementioned barriers and incentives, for example, the identified lack of influence between EU waste legislation and the ecodesign of products. In addition, the (proposed) thresholds in EU chemicals legislation and the difficulty to measure such refined concentrations of hazardous substances may affect the recycling of WEEE, more specifically, WEEE plastics. As compliance with these thresholds cannot be proven, interviewees argued that this will eventually lead to an impossibility to recycle WEEE plastics altogether. In turn, this will have negative consequences for meeting the recycling targets set by the WEEE Directive, as EEE increasingly consists of plastic components. In addition, the interaction between EU chemicals legislation and some of the proposed product requirements—i.e., the requirements on the use of hazardous substances or the use of recycled content, more specifically plastics—risks overlap between the Ecodesign Directive and the RoHS Directive and the REACH Regulation [52,82], respectively, leads to concerns about the compatibility with requirements in EU chemicals legislation, according to interviewees [89]. At the same time, the interaction between the three areas of legislation also provides opportunities, in the sense that it could lead to synergies. Ecodesign requirements could contribute to improving the link with the end-of-life stage and EU waste legislation: requirements on disassembly could be used to support and complement the requirements on disassembly in the WEEE Directive, and product requirements could be used to anticipate the achievement of potential new targets in the WEEE Directive or to ban materials that negatively influence reuse and recycling. More generally, synergies could be used to strengthen compliance and enforcement of existing legislation [6,52,89,109].

Third, not only could the interaction between and within the EU chemicals, product and waste legislation governing EEE be improved, but the legislation could also be better aligned with CE objectives [12]. It appeared from the literature, and it was also stated by several interviewees, that the current legal framework does not match CE ambitions or even hampers them. For example, the transition in the EEE chain is, according to interviewees, hampered by the issues related to the waste status of products and EU chemicals legislation, particularly with regard to the use of (secondary) plastics. Moreover, a disconnection between the current legislation and the CE objectives can be found in the waste legislation governing EEE. The obligations of the WEEE Directive and WFD may be in line with the CE and life cycle thinking, but there is a lack of focus on higher steps of the waste hierarchy in its actual provisions. For instance, reuse does not form a major part of the legal framework [6], the targets of the WEEE Directive are said to be stimulating recycling over reuse, and EPR is said to disadvantage circular companies. This shows that it is not only important that the objectives of legislation are in line with the CE, but also that its contents are to actually fulfil or realize those objectives.

The same counts for existing instruments and possibilities in legislation. The 65% collection target as well as the instrument of EPR are, moreover, criticized for only working moderately. In the Netherlands, there has been an increasing failure to achieve the 65% target in 2021 (see Section 2). Although this raises the question whether that is due to the functioning of the target itself or due to the implementation by the Netherlands, the fact that only three MSs did achieve the collection target seems to indicate an underlying issue [105]. In other words, it appears that just setting the target is pointless and that more is required to achieve the desired results. In addition, EPR is criticized for not meeting its objectives: it only has minimal effect on the design of products and is not sufficient for guaranteeing sustainable waste management. Lastly, the numerous examples regarding the potential of ecodesign Directive has been insufficiently utilized to date. All in all, the current legislation could be better aligned with the CE objectives and the current instruments and possibilities within the legislation could be better utilized to contribute to the CE transition in the EEE chain.

In summary, the results from the literature and empirical study seem to indicate that, next to the more practical matters, the legal barriers and incentives are mostly related to either the interaction within or between the three legal fields and respective legal acts, or a disconnection between the legislation and CE objectives. This is not entirely unexpected, as the current legislation to a large extent predates the start of the CE transition in the EU. The current legislation, therefore, has been designed on the basis of linear thinking, instead of circular and life cycle thinking (see also Section 3). In order to enable or stimulate the CE transition in the EEE chain, it might be desirable to adjust the EU chemicals, product and waste legislation governing EEE based on circular and life cycle thinking, while also taking into account the abovementioned practical aspects, such as the clarity and enforceability of legislation.

The results of the literature and interview study give reason to reflect on the approach within the EU chemicals, product and waste legislation governing EEE. In the literature, it has been suggested to adopt a holistic approach for (W)EEE management to account for the whole life cycle [6,69] and also to take into account economic, technical, sustainability and social aspects, as well as the cooperation between actors within the EEE supply chain. This could address the finding that there is a need to improve the interaction between the three legal fields as well as the alignment of the legislation with CE objectives. Adopting such a holistic approach in EU chemicals, product and waste legislation governing EEE would mean that life cycle thinking, including CE objectives, would underpin the legislation and that the legal framework would be shaped accordingly [69]. This would build on the already existing attention that is, in some cases, present in the (recitals of the) legal acts (see Section 3).

A holistic approach within EU chemicals, product and waste legislation governing EEE could, in line with the current approach of the legislation, focus on the sector of EEE. However, in the interviews it was discussed whether the differences within the EEE sector could or should lead to a specific focus on (a number of) separate products, as is already the case with the ecodesign requirements and, to a more limited extent, with the product categories in the WEEE Directive [8,73]. Some interviewees mentioned that products within the EEE chain are all very different from each other and that therefore attention should be paid to the degree in which identified barriers and incentives apply to, respectively, have a positive CE impact on all EEE. In addition, these differences make it difficult to set generic requirements or targets for all EEE. On top of that, as multiple of the identified issues and incentives seem to be related to specific materials in (W)EEE, the question arose if instead of a (more) product-specific approach, a (more) material-specific approach might be beneficial to the CE transition in the EEE chain. As appears from literature, but also according to several interviewees, it often are the (hazardous substances in) (W)EEE plastics that lead

to barriers or require incentives regarding their recycling and reuse. The same counts for hazardous substances in general or CRMs [110]. This creates the impression that several barriers and incentives do not relate to specific products, but to specific materials.

Alongside the reflections on the approach within the EU chemicals, product and waste legislation governing EEE, in particular the finding that there is a need for improving the interaction and alignment with CE objectives, underlines the importance and difficulty of finding a balance between interests that appear difficult to reconcile. For example, between stimulating the CE transition by promoting reuse or recycling on the one hand, and ensuring the safety of human health and the environment on the other hand [63]. Especially the identified barriers and incentives that relate to EU chemicals legislation and (W)EEE plastics show that safeguarding human health and the environment could have consequences for more circularity, and vice versa [111]. Similar considerations play a role in choices regarding life time extension versus energy efficiency [52,112], modularity versus higher environmental impact or [113], according to interviewees, repairability and consumer-safety. The difficulty and the necessity of finding a balance between such interests not only plays a role in the EU chemicals, product and waste legislation governing EEE, but also in relation to the concept of the CE in general [110]. Furthermore, in both the literature and interviews, it was noted that too much and too detailed legislation, particularly product requirements, could impede innovation [6,95], which would be undesirable in view of the transition towards CE. On the other hand, many interviewees were positive about more legislation. As the interviews revealed that actors regularly point at and wait for each other when it comes to taking steps towards a CE, legislation could be used to get the sector moving and to 'push' actors, especially producers, in the right direction.

## 5.1. Current and Future Developments

The overview of barriers and incentives has shown that there are still many legal barriers that could be taken away and many incentives that could be implemented in order to facilitate the transition towards a CE in the EEE chain. These identified barriers and (lack of) incentives are related to both the EU chemicals, product and waste legislation itself and to the implementation or approach by the MSs, in this case, the Netherlands. This means that with regard to the EU chemicals, product and waste legislation governing EEE barriers could be taken away and incentives could be provided at both the EU level and the Dutch level. In some cases, there might already be possibilities for MSs to take away barriers or provide incentives, i.e., to pick the low-hanging fruit. An example is the possibility in Article 8a WFD to introduce modulated fees to therewith promote circular businesses and products, despite the divided opinions of the interviewees. There are also examples where a MS, in this case the Netherlands, takes or has taken action itself, such as with regard to the afgifteplicht or the AVV. Many barriers and incentives, however, would require action at the EU level, such as reviewing target references in the WEEE Directive or to introduce more circular ecodesign requirements. As already stated in the introduction, amendments of EU chemicals, product and waste legislation have already taken place since the official start of the transition towards a CE, such as the amendments of the WEEE Directive and WFD in 2018. Further amendments and reviews are planned for the near future.

In March 2022, the EC presented the proposal for the Ecodesign for Sustainable Products Regulation, which has as its objective 'to reduce the negative life cycle environmental impacts of products and improve the functioning of the internal market', and will apply to almost all physical products on the EU market. It proposes a framework for setting product requirements in product specific legislation on energy efficiency as well as on sustainability and circularity aspects, such as durability, resource efficiency or recycled content [29]. In addition, the Regulation proposes new information requirements, including digital product passports. These passports should not only help consumers to make informed choices, but also provide access to relevant information to recyclers and repairers, and contribute to improving the enforcement of legal requirements. Labelling requirements, including the incorporation of circularity aspects in energy labels, as well as the introduction of a new Ecodesign for Sustainable Products label, have also been proposed [29]. Until the proposal of the Ecodesign Regulation is adopted, there will be an increased emphasis on integrating circular economy aspects in implementing regulations based on the current Ecodesign Directive [114].

In several aspects, the proposal for the Ecodesign Regulation corresponds to the identified incentives, particularly with regard to the introduction of circular ecodesign requirements and the use of labels, but also with regard to strengthening the level playing field in the EU. Regarding the digital product passport, it is remarkable that multiple interviewees, among whom waste treatment operators, a refurbisher, and producer (organizations), were very critical. While some interviewees admitted that the information stored in such passports could be useful as an overall information tool for within the EEE supply chain, waste treatment operators and refurbishers specifically stated that passports with repair or treatment information do not fit their current processes and that they themselves would therefore not consult such passports.

The CEAP 2020 also announced a Circular Electronics Initiative, which inter alia includes a focus on decreasing the environmental impact of mobile phones and tablets [115], the right to repair and regulatory measures on chargers for mobile phones and similar devices [3]. In addition, the RoHS Directive is being revised to address several practical and systemic issues, such as high complexity and administrative burden, difficulties regarding enforcement, insufficient provisions to support the CE and consistency with the REACH Regulation and Ecodesign Directive [116]. The proposal for the revised Directive is expected in the last quarter of 2022 [117]. Following the CEAP and EU Green Deal, the WSR has also been revised, which has resulted in a proposal for a new Regulation. The objectives of the revision were to facilitate waste shipments in the EU, to align the WSR with CE objectives, to tackle illegal waste shipments and to ensure that waste exports to third countries are managed in an environmentally sound way. Amongst other things, changes have been proposed to the procedure for waste shipments in order to reduce administrative burdens, waste exports to third countries shall require evidence of environmentally sound treatment, and measures are proposed to improve the enforcement of the WSR [118]. This seems to be largely in line with the identified barriers and incentives relating to the WSR (see Section 4.3). Furthermore, the REACH Regulation and CLP Regulation are being revised [119,120], the EC is currently working on a revision of the WFD [121], and an evaluation of the WEEE Directive has recently been announced [122].

## 6. Conclusions

In light of the transition towards a more circular EEE chain, this article discusses legal barriers and incentives with regard to the EU chemicals, product and waste legislation governing EEE as well as the implementation of this legislation in the Netherlands. These legal barriers and incentives were identified in literature and in interviews with practitioners in the Netherlands, in which the results of the literature study were largely confirmed. The results show that changes to the EU chemicals, product and waste legislation governing EEE are necessary to take away barriers and incentivize the transition towards a more circular EEE chain. It would be desirable to improve the interaction within the legal framework governing the life cycle of EEE, as this interaction leads to (potential) issues, but also because this interaction would provide opportunities in creating synergies. It would also be beneficial to better align the legislation with the objectives of the CE and life cycle thinking, including shifting the focus of the WEEE Directive and WFD to higher levels of the waste hierarchy, as well as making better use of existing instruments and possibilities in legislation, such as improving EPR and setting more circular ecodesign requirements. This, without losing sight of practical aspects, such as the enforceability and complexity of the legislation. The numerous future developments at the EU level, especially the proposal for the Ecodesign for Sustainable Products Regulation, seem promising in this regard. However, attention should also be paid to the already existing possibilities for MSs to take action with regard to the legal barriers and incentives at the national level.

Overall, the identified legal barriers and incentives give reason to argue that a holistic approach within the EU chemicals, product and waste legislation governing EEE, reflecting the whole life cycle of EEE, would benefit the transition towards a more circular EEE chain. While such a holistic approach could be applied within the current sector-specific approach of EU chemicals, product and waste legislation governing EEE, this research additionally shows that a more product-specific or even material-specific approach could be more beneficial. Further research on this is needed. The same counts for the question on how to deal with the legal solutions and incentives that seem irreconcilable or require balancing contradictory interests, such as increasing reuse and recycling on the one hand and protecting human health and the environment on the other.

In short, the identified legal barriers and incentives that are related to the EU chemicals, product and waste legislation governing EEE indicate that changes to the legal framework are necessary for the transition towards a more circular EEE chain. This article has given an overview of possibilities and suggestions to do so from a theoretical and empirical perspective, and therewith contributes to creating a legal framework that is conducive to the CE transition, both with regard to the EEE chain and in general.

The research in this article will be continued by similar research into two other product streams, in order to examine whether and to what extent the findings also apply to these, and in order to draw conclusions on the legal barriers and incentives for the CE transition in EU chemicals, product and waste legislation in general. Next to legal barriers and incentives, the literature and interview study identified multiple barriers and incentives that fell outside the scope of this research (see Section 3). Directions for further research, therefore, include looking into additional (non-legal) steps that might need to be taken to facilitate and stimulate the transition towards a more circular EEE chain [6], as well as the interaction of such additional steps with the EU chemicals, product and waste legislation governing EEE.

Funding: This research received no external funding.

**Institutional Review Board Statement:** The study was conducted in accordance with the Declaration of Helsinki, and approved by the Ethics Review Committee of the Faculty of Law, Economics and Governance of Utrecht University.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

**Data Availability Statement:** Restrictions apply to the availability of these data. Data was obtained from the interview participants and are only available from the author with the permission of the interview participants.

Conflicts of Interest: The author declares no conflict of interest.

## Appendix A. Questionnaire for the Semi-Structured Interviews

- 1. What is your experience with or opinion on the current chemicals, product and waste legislation in light of the transition towards a circular EEE chain?
- 2. Do you identify issues or barriers to the transition towards a more circular EEE chain?
- 3. What solutions do you think would be desirable to remove identified obstacles?
- 4. Are there incentives that, in your opinion, could stimulate the transition towards a circular EEE chain?
- 5. Looking at the goal of realizing a 50% circular economy by 2030 and a 100% circular economy by 2050, what do you consider as big challenges for the EEE chain and what should really be focused on?

# Appendix B

**Table A1.** Implementation of the EU chemicals, product and waste legislation governing EEE in the Netherlands.

EU Legislation	Implementation in The Netherlands		
RoHS Directive	Regulation of hazardous susbtances in electrical and electronical equipment ( <i>Regeling gevaarlijke stoffen in elektrische en elektronische apparatuur</i> )		
REACH Regulation	Chapter 9-Environmental Protection Act (Wet milieubeheer)		
CLP Regulation	Chapter 9-Environmental Protection Act		
Ecodesign Directive	Chapter 9-Environmental Protection Act		
Energy-labelling Regulation	Decree on energy-labelling energyrelated products ( <i>Besluit energie-etikettering energiegerelateerde producten</i> )		
WEEE Directive	Regulation of waste electrical and electronic equipment ( <i>Regeling afgedankte elektrische en elektronische apparatuur</i> )		
WFD	Chapter 10-Environmental Protection Act		

## References

- Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions Closing the Loop—An EU Action Plan for the Circular Economy. 2015. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX:52015DC0614 (accessed on 4 October 2022).
- Communication from the Commission to the European Parliament, the European Council, the Council, the European Economic and Social Committee and the Committee of the Regions the European Green Deal. 2019. Available online: https://eur-lex. europa.eu/legal-content/EN/TXT/?uri=COM%3A2019%3A640%3AFIN (accessed on 4 October 2022).
- Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions a New Circular Economy Action Plan for a Cleaner and More Competitive Europe. 2020. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=COM%3A2020%3A98%3AFIN (accessed on 4 October 2022).
- 4. Shittu, O.S.; Williams, I.D.; Shaw, P.J. Global E-waste management: Can WEEE make a difference? A review of e-waste trends, legislation, contemporary issues and future challenges. *Waste Manag.* 2021, *120*, 549–563. [CrossRef] [PubMed]
- 5. Zhang, S.; Ding, Y.; Liu, B.; Chang, C. Supply and demand of some critical metals and present status of their recycling in WEEE. *Waste Manag.* **2017**, *65*, 113–127. [CrossRef] [PubMed]
- 6. Cole, C.; Gnanapragasam, A.; Cooper, T.; Singh, J. An assessment of achievements of the WEEE Directive in promoting movement up the waste hierarchy: Experiences in the UK. *Waste Manag.* **2019**, *87*, 417–427. [CrossRef] [PubMed]
- Forti, V.; Baldé, C.P.; Kuehr, R.; Bel, G. *The Global E-Waste Monitor 2020: Quantities, Flows, and the Circular Economy Potential*; United Nations University (UNU)/United Nations Institute for Training and Research (UNITAR)—Co-Hosted SCYCLE Programme, International Telecommunication Union (ITU) & International Solid Waste Association (ISWA): Bonn, Germany; Geneva, Switzerland; Rotterdam, The Netherlands, 2020.
- 8. Directorate-General for Internal Market, Industry; Bobba, S.; Claudiu, P.; Huygens, D.; Alves Dias, P.; Gawlik, B.; Tzimas, E.; Wittmer, D.; Nuss, P.; Grohol, M.; et al. *Report on Critical Raw Materials and the Circular Economy*; Publications Office of the European Union: Maastricht, The Netherlands, 2018; ISBN 978-92-79-94626-4.
- Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions Critical Raw Materials Resilience: Charting a Path towards greater Security and Sustainability. 2020. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52020DC0474 (accessed on 4 October 2022).
- Commission Staff Working Document Measuring Progress towards Circular Economy in the European Union—Key Indicators for a Monitoring Framework Accompanying the Document Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions on a Monitoring Framework for the Circular Economy. 2018. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=SWD%3A2018%3A17% 3AFIN (accessed on 4 October 2022).
- 11. Bacher, J.; Dams, Y.; Duhoux, T.; Deng, Y.; Teittinen, T.; Mortensen, L.F. *Electronic Products and Obsolescence in a Circular Economy*; European Topic Centre Waste and Materials in a Green Economy: Copenhagen Denmark, 2020.
- Rizos, V.; Bryhn, J.; Alessi, M.; Righetti, E.; Fujiwara, N.; Stroia, C. Barriers and Enablers for Implementing Circular Economy Business Models: Evidence from the Electrical and Electronic Equipment and Agri-Food Value Chains; CEPS Research Report; CEPS: Brussels, Belgium, 2021; p. 64.

- Althaf, S.; Babbitt, C.W.; Chen, R. Forecasting electronic waste flows for effective circular economy planning. *Resour. Conserv. Recycl.* 2019, 151, 104362. [CrossRef]
- Kiddee, P.; Naidu, R.; Wong, M.H. Electronic waste management approaches: An overview. Waste Manag. 2013, 33, 1237–1250. [CrossRef]
- 15. Işıldar, A.; Rene, E.R.; van Hullebusch, E.D.; Lens, P.N.L. Electronic waste as a secondary source of critical metals: Management and recovery technologies. *Resour. Conserv. Recycl.* 2018, 135, 296–312. [CrossRef]
- 16. van der Veen, G. (Ed.) *Regulatory Barriers for the Circular Economy: Lessons from Ten Case Studies;* Technopolis Group: Brighton, UK, 2016.
- 17. Bressanelli, G.; Saccani, N.; Pigosso, D.C.A.; Perona, M. Circular Economy in the WEEE industry: A systematic literature review and a research agenda. *Sustain. Prod. Consum.* **2020**, *23*, 174–188. [CrossRef]
- 18. Rizos, V.; Bryhn, J. Implementation of circular economy approaches in the electrical and electronic equipment (EEE) sector: Barriers, enablers and policy insights. J. Clean. Prod. 2022, 338, 130617. [CrossRef]
- 19. de Römph, T.J.; Van Calster, G. REACH in a circular economy: The obstacles for plastics recyclers and regulators. *Rev. Eur. Comp. Int. Environ. Law* **2018**, *27*, 267–277. [CrossRef]
- 20. de Römph, T.J. *The Legal Transition towards a Circular Economy: EU Environmental Law Examined;* KU Leuven and Hasselt University: Hasselt, Belgium, 2018.
- 21. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions on the Implementation of the Circular Economy Package: Options to Address the Interface between Chemical, Product and Waste Legislation (Text with EEA Relevance) Options to Address the Interface between Chemical, Product and Waste Legislation. 2018. Available online: https://eur-lex.europa.eu/legal-content/en/TXT/?uri=CELEX%3A52018DC0032 (accessed on 4 October 2022).
- Directive 2011/65/EU of the European Parliament and of the Council of 8 June 2011 on the Restriction of the Use of Certain Hazardous Substances in Electrical and Electronic Equipment (Recast) Text with EEA Relevance. 2011, Volume 174. Available online: https://eur-lex.europa.eu/eli/dir/2011/65/oj/eng (accessed on 17 October 2022).
- Directive 2012/19/EU of the European Parliament and of the Council of 4 July 2012 on Waste Electrical and Electronic Equipment (WEEE) (Recast) (Text with EEA Relevance). 2018. Available online: https://eur-lex.europa.eu/eli/dir/2012/19/oj/eng (accessed on 17 October 2022).
- 24. Directive 2009/125/EC of the European Parliament and of the Council of 21 October 2009 Establishing a Framework for the Setting of Ecodesign Requirements for Energy-Related Products (Recast) (Text with EEA Relevance). 2009. Available online: https://eur-lex.europa.eu/eli/dir/2009/125/oj/eng (accessed on 17 October 2022).
- Directive 2008/98/EC of the European Parliament and of the Council of 19 November 2008 on Waste and Repealing Certain Directives (Text with EEA Relevance). 2018. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX% 3A32008L0098 (accessed on 17 October 2022).
- Directive (EU) 2018/851 of the European Parliament and of the Council of 30 May 2018 Amending Directive 2008/98/EC on waste (Text with EEA Relevance). 2018, Volume 150. Available online: https://eur-lex.europa.eu/eli/dir/2018/851/oj/eng (accessed on 4 October 2022).
- Directive (EU) 2018/849 of the European Parliament and of the Council of 30 May 2018 Amending Directives 2000/53/EC on End-of-Life Vehicles, 2006/66/EC on Batteries and Accumulators and Waste Batteries and Accumulators, and 2012/19/EU on Waste Electrical and Electronic Equipment (Text with EEA Relevance). 2018, Volume 150. Available online: https://eur-lex. europa.eu/eli/dir/2018/849/oj/eng (accessed on 4 October 2022).
- 28. Directive (EU) 2017/2102 of the European Parliament and of the Council of 15 November 2017 Amending Directive 2011/65/EU on the Restriction of the Use of Certain Hazardous Substances in Electrical and Electronic Equipment (Text with EEA Relevance). 2017, Volume 305. Available online: https://eur-lex.europa.eu/eli/dir/2017/2102/oj/eng (accessed on 4 October 2022).
- Proposal for a Regulation of the European Parliament and of the Council Establishing a Framework for Setting Ecodesign Requirements for Sustainable Products and Repealing Directive 2009/125/EC. 2022. Available online: <a href="https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=CELEX:52022PC0142">https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=CELEX:52022PC0142</a> (accessed on 4 October 2022).
- de Waal, I.M. Coherence in law: A way to stimulate the transition towards a circular economy? A critical analysis of the European Commission's aspiration to achieve full coherence between chemicals legislation and waste legislation—And product legislation. *Maastricht J. Eur. Comp. Law* 2021, 28, 760–783. [CrossRef]
- 31. Patton, M.Q. Qualitative Research & Evaluation Methods; SAGE: Newcastle upon Tyne, UK, 2002; ISBN 978-0-7619-1971-1.
- Palinkas, L.A.; Horwitz, S.M.; Green, C.A.; Wisdom, J.P.; Duan, N.; Hoagwood, K. Purposeful Sampling for Qualitative Data Collection and Analysis in Mixed Method Implementation Research. *Adm. Policy Ment. Health Ment. Health Serv. Res.* 2015, 42, 533–544. [CrossRef] [PubMed]
- 33. Boeije, H.R. Analysis in Qualitative Research; SAGE: Newcastle upon Tyne, UK, 2009; ISBN 978-1-4462-4135-6.
- 34. Marshall, M.N. Sampling for qualitative research. Fam. Pract. 1996, 13, 522–526. [CrossRef]
- 35. Webley, L. Qualitative Approaches to Empirical Legal Research; Oxford University Press: Oxford, UK, 2010.
- 36. Maguire, M.; Delahunt, B. Doing a Thematic Analysis: A Practical, Step-by-Step Guide for Learning and Teaching Scholars. *All Irel. J. High. Educ.* **2017**, *8*, 14.

- Centraal Bureau voor de Statistiek, Voor de Kleinere Grondstofvoetafdruk, Meer Recycling dan Gemiddeld in de EU. Available online: https://www.cbs.nl/nl-nl/nieuws/2020/08/kleinere-grondstofvoetafdruk-meer-recycling-dan-gemiddeld-in-de-eu (accessed on 3 January 2021).
- 38. Waste Statistics. Available online: https://ec.europa.eu/eurostat/statistics-explained/index.php?title=Waste\_statistics (accessed on 4 October 2022).
- 39. Statistics | Eurostat. Available online: https://ec.europa.eu/eurostat/databrowser/view/T2020\_RT130/bookmark/table?lang= en&bookmarkId=a69be825-957e-473c-a81f-f02866dc9141 (accessed on 16 September 2022).
- Rapportage 2018. Nationaal (W)EEE Register, 2019. Available online: https://www.nationaalweeeregister.nl/assets/uploads/ PDF/2018/2020-04-20%20Rapportage%20Nationaal%20(W)EEE%20Register%202018%20-%20def%20-%20v2.pdf (accessed on 20 November 2022).
- Rapportage 2019. Nationaal (W)EEE Register, 2020. Available online: https://www.nationaalweeeregister.nl/assets/uploads/ PDF/2020/Nationaal%20(W)EEE%20Register%20-%20rapportage%202019.pdf (accessed on 20 November 2022).
- Rapportage 2020. Nationaal (W)EEE Register, 2021. Available online: https://www.nationaalweeeregister.nl/assets/uploads/ PDF/2021/Rapportage%202020%20def%2020210628.pdf (accessed on 20 November 2022).
- Rapportage 2021. Nationaal (W)EEE Register, 2022. Available online: https://www.nationaalweeeregister.nl/assets/uploads/ PDF/2022/Rapportage%202021\_V20220627.pdf (accessed on 20 November 2022).
- 44. Koninkrijksrelaties, M.; Van, B.Z. en Regeling Afgedankte Elektrische en Elektronische Apparatuur. Available online: https://wetten.overheid.nl/BWBR0034782/2020-12-10 (accessed on 20 November 2022).
- 45. Regulation (EC) No 1907/2006 of the European Parliament and of the Council of 18 December 2006 concerning the Registration, Evaluation, Authorisation and Restriction of Chemicals (REACH), Establishing a European Chemicals Agency, Amending Directive 1999/45/EC and repealing Council Regulation (EEC) No 793/93 and Commission Regulation (EC) No 1488/94 as well as Council Directive 76/769/EEC and Commission Directives 91/155/EEC, 93/67/EEC, 93/105/EC and 2000/21/EC (Text with EEA Relevance)Text with EEA Relevance. 2022. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32006R1907&cgid=1671814503944 (accessed on 17 October 2022).
- 46. Regulation (EC) No 1272/2008 of the European Parliament and of the Council of 16 December 2008 on Classification, Labelling and Packaging of Substances and Mixtures, Amending and Repealing Directives 67/548/EEC and 1999/45/EC, and Amending Regulation (EC) No 1907/2006 (Text with EEA Relevance) Text with EEA Relevance. 2022. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32008R1272&qid=1671814707327 (accessed on 17 October 2022).
- 47. Regulation (EU) 2017/1369 of the European Parliament and of the Council of 4 July 2017 Setting a Framework for Energy Labelling and Repealing Directive 2010/30/EU (Text with EEA Relevance) Text with EEA Relevance. 2021. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32017R1369&qid=1671814770446 (accessed on 17 October 2022).
- Regulation (EC) No 1013/2006 of the European Parliament and of the Council of 14 June 2006 on Shipments of Waste. 2021. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32006R1013&qid=1671814792696 (accessed on 17 October 2022).
- Bugge, H.C.; Dalhammar, C.; Maitre-Ekern, E. Developing legislation to prevent environmental damage from products: A herculean but necessary task. In *Preventing Environmental Damage from Products*; Cambridge University Press: Cambridge, UK, 2018; pp. 1–20. ISBN 978-1-108-42244-4.
- de Römph, T.J. Pressing Forward—Developments in the Transition Towards Sustainable Materials Management in EU Environmental Law. In *Legal Aspects of Sustainable Development*; Mauerhofer, V., Ed.; Springer International Publishing: Cham, Switzerland, 2016; pp. 511–532. ISBN 978-3-319-26019-8.
- 51. Turunen, T. The Concepts of Waste and Non-Waste in the Circular Economy; Itä-Suomen Yliopisto: Kuopio, Finland, 2018; ISBN 978-952-61-2920-4.
- 52. Bundgaard, A.M.; Remmen, A.; Zacho, K.O. *Ecodesign Directive Version 2.0: From Energy Efficiency to Resource Efficiency;* Miljøstyrelsen: København, Danmark, 2015.
- Commission Staff Working Document Sustainable Products in a Circular Economy—Towards an EU Product Policy Framework Contributing to the Circular Economy. 2019. Available online: https://ec.europa.eu/environment/pdf/circular-economy/ sustainable\_products\_circular\_economy.pdf (accessed on 6 October 2022).
- 54. Mathieux, F.; Ardente, F.; Bobba, S. Ten years of scientific support for integrating circular economy requirements in the EU ecodesign directive: Overview and lessons learnt. *Procedia CIRP* **2020**, *90*, 137–142. [CrossRef]
- 55. Commission Implementing Regulation (EU) 2015/341 of 20 February 2015 Laying down Detailed Rules for Implementing Regulation (EU) No 223/2014 of the European Parliament and of the Council as Regards the Models for Submission of Certain Information to the Commission. 2015, Volume 60. Available online: https://eur-lex.europa.eu/eli/reg\_impl/2015/341/oj/eng (accessed on 20 November 2022).
- 56. Tesar, M.; Karigl, B.; Lampert, C.; Neubauer, C.; Oliva, J.; Wolf, J. Study on Quality Standards for the Treatment of Waste Electrical and Electronic Equipment (WEEE): Final Report; European Commission, Directorate-General for Environment: Brussels, Belgium, 2021.

- 57. In't Groen, B.; Stengs, L.; Zonneveld, N. European Standards for Waste Electrical and Electronic Equipment (WEEE). 2017. Available online: https://www.cencenelec.eu/media/CEN-CENELEC/AreasOfWork/CEN-CENELEC\_Topics/Environment%20 and%20Sustainability/Quicklinks%20General/Documentation%20and%20Materials/weee-brochure.pdf (accessed on 7 October 2022).
- Maitre-Ekern, E. The Choice of Regulatory Instruments for a Circular Economy. In *Environmental Law and Economics*; Mathis, K., Huber, B.R., Eds.; Springer International Publishing: Cham, Switzerland, 2017; pp. 305–334. ISBN 978-3-319-50931-0.
- 59. Proposal for a Regulation of the European Parliament and of the Council on Shipments of Waste. 2003. Available online: https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=COM%3A2003%3A379%3AFIN (accessed on 20 November 2022).
- 60. den Heijer, M.; van Os van den Abeelen, T.; Maslyka, A. *On the Use and Misuse of Recitals in European Union Law*; Amsterdam Center for International Law No. 2019-15; Amsterdam Center for International Law: Amsterdam, The Netherlands, 2019.
- 61. Report from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions on the Implementation of the Circular Economy Action Plan. 2017. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52017DC0033 (accessed on 4 October 2022).
- 62. Maitre-Ekern, E.; Dalhammar, C. Regulating Planned Obsolescence: A Review of Legal Approaches to Increase Product Durability and Reparability in Europe. *Rev. Eur. Comp. Int. Environ. Law* **2016**, *25*, 378–394. [CrossRef]
- 63. Dalhammar, C.; Milios, L.; Richter, J.L. *Increasing the Lifespan of Products: Policies and Consumer Perspectives*; ER; Energimyndigheten: Stockholm, Sweden, 2021.
- 64. Jepsen, D.; Spengler, L.; Augsberg, L. *Delivering Resource-Efficient Products: How Ecodesign Can Drive a Circular Economy in Europe*; European Environmental Bureau: Brussels, Belgium, 2015.
- 65. Scruggs, C.E.; Nimpuno, N.; Moore, R.B.B. Improving information flow on chemicals in electronic products and E-waste to minimize negative consequences for health and the environment. *Resour. Conserv. Recycl.* **2016**, *113*, 149–164. [CrossRef]
- 66. McMahon, K.; Johnson, M.; Fitzpatrick, C. Enabling preparation for re-use of waste electrical and electronic equipment in Ireland: Lessons from other EU member states. *J. Clean. Prod.* **2019**, 232, 1005–1017. [CrossRef]
- 67. Bundgaard, A.M.; Mosgaard, M.A.; Remmen, A. From energy efficiency towards resource efficiency within the Ecodesign Directive. J. Clean. Prod. 2017, 144, 358–374. [CrossRef]
- 68. Dalhammar, C. The Application of 'life cycle thinking' in European environmental law: Theory and practice. J. Eur. Environ. Plan. Law 2015, 12, 97–127. [CrossRef]
- 69. Friege, H. Review of material recovery from used electric and electronic equipment-alternative options for resource conservation. *Waste Manag. Res. J. Sustain. Circ. Econ.* **2012**, *30*, 3–16. [CrossRef]
- Golsteijn, L.; Valencia Martinez, E. The Circular Economy of E-Waste in the Netherlands: Optimizing Material Recycling and Energy Recovery. J. Eng. 2017, 2017, 1–6. [CrossRef]
- Tecchio, P.; Ardente, F.; Marwede, M.; Christian, C.; Dimitrova, G.; Mathieux, F. Analysis of Material Efficiency Aspects of Personal Computers Product Group, EUR 28394 EN; Publications Office of the European Union: Luxembourg, 2018.
- 72. Polverini, D. Regulating the circular economy within the ecodesign directive: Progress so far, methodological challenges and outlook. *Sustain. Prod. Consum.* **2021**, *27*, 1113–1123. [CrossRef]
- 73. Mazahir, S.; Verter, V.; Boyaci, T.; Van Wassenhove, L.N. Did Europe Move in the Right Direction on E-waste Legislation? *Prod. Oper. Manag.* **2019**, *28*, 121–139. [CrossRef]
- 74. Balde, C.P.; van den Brink, S.; Forti, V.; van der Schalk, A.; Hopstoken, F. The Dutch WEEE Flows 2020: What happened between 2010 and 2018? 2020. Available online: https://www.scycle.info/the-dutch-weee-flows-2020-what-happened-between-2010 -and-2018/ (accessed on 4 October 2022).
- 75. Commission Notice the 'Blue Guide' on the Implementation of EU Product Rules 2022 (Text with EEA Relevance) 2022/C 247/01. 2022. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=uriserv%3AOJ.C\_.2022.247.01.0001.01.ENG& toc=OJ%3AC%3A2022%3A247%3ATOC (accessed on 4 October 2022).
- 76. Commission Staff Working Document Accompanying the Document Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions on the Implementation of the Circular Economy Package: Options to Address the Interface between Chemical, Product and Waste Legislation. 2018. Available online: https://eur-lex.europa.eu/legal-content/en/TXT/?uri=CELEX:52018SC0020 (accessed on 4 October 2022).
- 77. Commission Staff Working Document Fitness Check of the Most Relevant Chemicals Legislation (Excluding REACH), as well as Related Aspects of Legislation Applied to Downstream Industries Accompanying the Document Report from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions Findings of the Fitness Check of the Most Relevant Chemicals Legislation (Excluding REACH) and Identified Challenges, Gaps and Weaknesses. 2019. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=SWD:2019: 199:FIN&qid=1561480217001 (accessed on 4 October 2022).
- 78. Bernard, A. Buonsante Vito Keeping It Clean: How to Protect the Circular Economy from Hazardous Substances; European Environmental Bureau: Brussels, Belgium, 2017.
- Council and Parliament Agree to Reduce Limit Values for the Presence of Persistent Organic Pollutants in Waste. Available online: https://www.consilium.europa.eu/en/press/press-releases/2022/06/21/council-and-parliament-agree-to-reduce-limitvalues-for-the-presence-of-persistent-organic-pollutants-in-waste/ (accessed on 4 October 2022).

- 80. Ardente, F.; Mathieux, F. Environmental assessment of the durability of energy-using products: Method and application. *J. Clean. Prod.* **2014**, *74*, 62–73. [CrossRef]
- European Environmental Agency. Europe's Consumption in a Circular Economy: The Benefits of Longer-Lasting Electronics. 2020. Available online: https://www.eea.europa.eu/publications/europe2019s-consumption-in-a-circular/benefits-of-longer-lasting-electronics (accessed on 6 October 2022).
- Ardente, F.; Mathieux, F. Identification and assessment of product's measures to improve resource efficiency: The case-study of an Energy using Product. J. Clean. Prod. 2014, 83, 126–141. [CrossRef]
- 83. Tecchio, P.; McAlister, C.; Mathieux, F.; Ardente, F. In search of standards to support circularity in product policies: A systematic approach. J. Clean. Prod. 2017, 168, 1533–1546. [CrossRef]
- Ardente, F.; Mathieux, F.; Recchioni, M. Recycling of electronic displays: Analysis of pre-processing and potential ecodesign improvements. *Resour. Conserv. Recycl.* 2014, 92, 158–171. [CrossRef]
- 85. Len, M. Improving Product Reparability: Policy Options at EU Level; RREUSE: Brussels, Belgium, 2015.
- 86. Vanegas, P.; Peeters, J.R.; Cattrysse, D.; Tecchio, P.; Ardente, F.; Mathieux, F.; Dewulf, W.; Duflou, J.R. Ease of disassembly of products to support circular economy strategies. *Resour. Conserv. Recycl.* **2018**, 135, 323–334. [CrossRef]
- 87. Mathieux, F.; Recchioni, M.; Ardente, F. Measuring the Time for Extracting Components in End-of-life Products: Needs for a Standardized Method and Aspects to be Considered. *Procedia CIRP* **2014**, *15*, 245–250. [CrossRef]
- Talens Peiró, L.; Polverini, D.; Ardente, F.; Mathieux, F. Advances towards circular economy policies in the EU: The new Ecodesign regulation of enterprise servers. *Resour. Conserv. Recycl.* 2020, 154, 104426. [CrossRef]
- Dalhammar, C. Industry attitudes towards ecodesign standards for improved resource efficiency. J. Clean. Prod. 2016, 123, 155–166. [CrossRef]
- Commission Regulation (EU) 2019/424 of 15 March 2019 Laying down Ecodesign Requirements for Servers and Data Storage Products Pursuant to Directive 2009/125/EC of the European Parliament and of the Council and Amending Commission Regulation (EU) No 617/2013 (Text with EEA Relevance). 2019, Volume 074. Available online: https://eur-lex.europa.eu/eli/ reg/2019/424/oj/eng (accessed on 4 October 2022).
- 91. Commission Regulation (EU) 2019/1783 of 1 October 2019 amending Regulation (EU) No 548/2014 on Implementing Directive 2009/125/EC of the European Parliament and of the Council with Regard to Small, Medium and Large Power Transformers (Text with EEA Relevance). 2019, Volume 272. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=uriserv% 3AOJ.L\_2019.272.01.0074.01.ENG&toc=OJ%3AL%3A2019%3A272%3ATOC (accessed on 20 November 2022).
- 92. Kemna, R. *Methodology for Ecodesign of Energy-Related Products MEErP 2011;* European Commission, DG Enterprise and Industry: Brussels, Belgium, 2014.
- 93. Polverini, D.; Miretti, U. An approach for the techno-economic assessment of circular economy requirements under the Ecodesign Directive. *Resour. Conserv. Recycl.* 2019, 150, 104425. [CrossRef]
- Tecchio, P.; Ardente, F.; Marwede, M.; Clemm, C.; Dimitrova, G.; Mathieux, F. Ecodesign of Personal Computers: An Analysis of the Potentials of Material Efficiency Options. *Procedia CIRP* 2018, 69, 716–721. [CrossRef]
- 95. Backes, C. Faciliteert het omgevingsrecht innovaties op weg naar een circulaire economie? In *Met Recht Naar Een Circulaire Economie*; VMR; Boom juridisch: Den Haag, The Netherlands, 2017.
- 96. Spaans, J. Circulaire economie: Meer dan minder afval alleen. In *Met Recht Naar Een Circulaire Economie*; VMR; Boom Juridisch: Den Haag, The Netherlands, 2017.
- Campbell-Johnston, K.; Roos Lindgreen, E.; de Waal, I.M.; Gulotta, T.M.; Mondello, G.; Salomone, R.; Vermeulen, W. Policy Brief on Critical Raw Materials and Their Integration in Extended Producer Responsibility and Eco-Design Policy. Available online: https://dspace.library.uu.nl/handle/1874/420795 (accessed on 4 October 2022).
- 98. European Commission. Directorate General for the Environment; IEEP; BIO Intelligence Service; Ecologic; psi; IVM. Scoping Study to Identify Potential Circular Economy Actions, Priority Sectors, Material Flows and Value Chains: Final Report; European Commission: Brussels, Belgium, 2014.
- Call to Revise EU Legislation for Waste Electrical and Electronic Equipment. ECOS. Available online: https://ecostandard.org/ news\_events/call-to-revise-eu-legislation-for-waste-electrical-and-electronic-equipment/ (accessed on 20 November 2022).
- 100. Naar een Economie zonder afval Brede maatschappelijke heroverweging; Rijksoverheid—Inspectie der Rijksfinanciën. 2020. Available online: https://open.overheid.nl/repository/ronl-c049c2ec-8376-4156-927e-70a7c120112d/1/pdf/bmh-11-naar-eeneconomie-zonder-afval.pdf (accessed on 8 December 2021).
- 101. Hanemaaijer, A.; Kishna, M.; Brink, H.; Koch, J.; Prins, A.G.; Rood, T. *Integrale Circulaire Economie Rapportage 2021*; PBL: Den Haag, The Netherlands, 2021.
- 102. Texts Adopted—New Circular Economy Action Plan—Wednesday. 10 February 2021. Available online: https://www.europarl.europa.eu/doceo/document/TA-9-2021-0040\_EN.html (accessed on 4 October 2022).
- 103. Kort, M.; van Grootel, M. Resultatenonderzoek: Plan van Aanpak 65% AEEA; Rebel: Rotterdam, The Netherlands, 2020.
- 104. Dimitropoulos, A.; Tijm, J.; in 't Veld, D. *Extended Producer Responsibility: Design, Functioning and Effects*; PBL and CPB: Den Haag, The Netherlands, 2021.
- 105. Waste Statistics—Electrical and Electronic Equipment. Available online: https://ec.europa.eu/eurostat/statistics-explained/ index.php?title=Waste\_statistics\_-\_electrical\_and\_electronic\_equipment (accessed on 4 October 2022).

- 106. Vermeulen, W.; Backes, C.; de Munck, M.; Campbell-Johnston, K.; de Waal, I.M.; Rosales Carreon, J.; Boeve, M. WHITE PAPER on Pathways for Extended Producer Responsibility on the road to a Circular Economy; Utrecht University, Circular Economy and Society Hub: Utrecht, The Netherlands, 2021.
- 107. Backes, C.; Boeve, M. Uitgebreide Producentenverantwoordelijkheid: Rol en Verantwoordelijkheden Gemeenten en Inzamelingsorganisaties in Relatie tot Producenten; UCWOSL Utrecht University: Utrecht, The Netherlands, 2022.
- Bakker, C.; Wang, F.; Huisman, J.; den Hollander, M. Products that go round: Exploring product life extension through design. J. Clean. Prod. 2014, 69, 10–16. [CrossRef]
- Ardente, F.; Calero Pastor, M.; Mathieux, F.; Talens Peiró, L. Analysis of end-of-life treatments of commercial refrigerating appliances: Bridging product and waste policies. *Resour. Conserv. Recycl.* 2015, 101, 42–52. [CrossRef]
- Alaranta, J.; Turunen, T. How to Reach a Safe Circular Economy?—Perspectives on Reconciling the Waste, Product and Chemicals Regulation. J. Environ. Law 2021, 33, 113–136. [CrossRef]
- 111. Barouta, D.; Alassali, A.; Picuno, C.; Bruno, M.; Syranidou, E.; Fiore, S.; Kuchta, K. E-plastics in a circular economy: A comprehensive regulatory review. J. Clean. Prod. 2022, 355, 131711. [CrossRef]
- 112. Bobba, S.; Ardente, F.; Mathieux, F. Environmental and economic assessment of durability of energy-using products: Method and application to a case-study vacuum cleaner. *J. Clean. Prod.* **2016**, *137*, 762–776. [CrossRef]
- 113. Schischke, K.; Proske, M.; Nissen, N.F.; Schneider-Ramelow, M. Impact of modularity as a circular design strategy on materials use for smart mobile devices. *MRS Energy Sustain*. **2019**, *6*, 16. [CrossRef]
- 114. Communication from the Commission Ecodesign and Energy Labelling Working Plan 2022-2024 2022/C 182/01. 2022. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX:52022XC0504(01) (accessed on 6 October 2022).
- 115. Inception Impact Assessment, 'Environmental Impact of Mobile Phones and Tablets'. Available online: https://ec.europa. eu/info/law/better-regulation/have-your-say/initiatives/12797-Designing-mobile-phones-and-tablets-to-be-sustainableecodesign\_en (accessed on 20 November 2022).
- Call for Evidence for an Impact Assessment, 'Review: Restriction of the Use of Hazardous Substances in Electronics'. Available online: https://ec.europa.eu/info/law/better-regulation/have-your-say/initiatives/13137-Review-Restriction-of-the-use-ofhazardous-substances-in-electronics\_en (accessed on 20 November 2022).
- 117. Parliament, E. Revision of EU Rules Restricting the Use of Hazardous Substances in Electronics (RoHS) (REFIT) | Legislative Train Schedule. Available online: https://www.europarl.europa.eu/legislative-train/theme-a-european-green-deal/file-revision-of-eu-rules-restricting-the-use-of-hazardous-substances-in-electronics (accessed on 1 September 2022).
- 118. Proposal for a Regulation of the European Parliament and of the Council on Shipments of Waste and Amending Regulations (EU) No 1257/2013 and (EU) No 2020/1056. 2021. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52021PC0709 (accessed on 4 October 2022).
- 119. Inception Impact Assessment, 'Revision of EU Legislation on Registration, Evaluation, Authorisation and Restriction of Chemicals'. Available online: https://ec.europa.eu/info/law/better-regulation/have-your-say/initiatives/12959-Chemicals-legislation-revision-of-REACH-Regulation-to-help-achieve-a-toxic-free-environment\_en (accessed on 20 November 2022).
- Inception Impact Assessment, 'Revision of EU Legislation on Hazard Classification, Labelling and Packaging of Chemicals'. Available online: https://ec.europa.eu/info/law/better-regulation/have-your-say/initiatives/12975-Revision-of-EU-legislation-on-hazard-classification-labelling-and-packaging-of-chemicals\_en (accessed on 20 November 2022).
- 121. Call for Evidence for an Impact Assessment, 'Environmental Impact of Waste Management—Revision of EU Waste Framework'. Available online: https://ec.europa.eu/info/law/better-regulation/have-your-say/initiatives/13225-Environmental-impactof-waste-management-revision-of-EU-waste-framework\_en (accessed on 20 November 2022).
- 122. Call for Evidence for an Evaluation 'Waste Electrical and Electronic Equipment—Evaluating the EU Rules'. Available online: https://ec.europa.eu/info/law/better-regulation/have-your-say/initiatives/13420-Waste-from-electrical-and-electronicequipment-evaluating-the-EU-rules\_en (accessed on 20 November 2022).

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.